Isaac Situation Report for Activities Conducted on 9/17/2012:
Stolthaven Update- LDEQ Staff:
UNITED STATES COAST GUARD (USCG) & LOUISIANA STATE POLICE (LSP) HAVE DETERMINED THE RESPONSE TO THIS INCIDENT IS NOW POST-EMERGENCY AS OF 1800 hrs ON SEPTEMBER 15, 2012. LDEQ IS CURRENTLY WORKING IN A UNIFIED COMMAND WITH THE USCG. 

Update from 9/17/12 activities:
1. A sheen was observed on a storm water discharge from the facility along the fence-line adjacent to Hwy 39.  The discharge flowed to roadside ditch along Hwy 39.  Facility reported the sheen to the LSP Hotline. According to Tim Smith with Stolthaven, the source of the sheen was not from a known product release, but from storm water from the facility as a result of the hurricane impact.   An Oil Spill Response Organization (OSRO) was dispatched to the area, and sorbent pads, sorbent boom, hard boom, and a vac truck were deployed to address the sheen.  Sandbags, rocks, and soil were placed in the areas of the overflow to reduce the flow of storm water offsite.  The discharge was sampled by CTEH to include pH, DEA, the full suite of analytes for NPDES Permits, and the parameters specified in the Offsite Sampling Plan. 
1. Tank Farms 
1.  Product Transfers
0. Tank B80-3 (DEA) is currently being transferred to frac tanks onsite.
0. Tank A50-7 (Ultra S4 lube oil) to be transferred to a barge beginning tonight
1. Skimming of oil from Tank Farm “A” containment into onsite frac tanks, which began on 9/10/12, is still ongoing. 80,500 gallons of oil recovered to date.   
1. Tanks E12.5-13 and E12.5-15 being used as “slop tanks” to store secondary containment contents.  Samples taken on 9/15/12 to determine DEA concentrations are currently being processed. 
1. Approximately 650,000 gallons wastewater from equalization Tank #1 (EQ-1) was transferred to Barge CGBM 109 and CGBM 10.  The transfer was completed today, 9/17/12.  The material will be shipped offsite to Stolthaven’s Houston facility for treatment/recovery and disposal. 
1. API inspections to check the external integrity of the tanks and lines are currently being conducted.
1. Offsite Drum Recovery and pre-storm onsite drum inventory.
2. This will continue to be a high priority.
2. Container recovery and recon within previously inaccessible areas within the community is ongoing
2. Recovered drums are being stored in the containment of the drum storage area.
1. Sample room cleanup is being conducted. Sample jars are being over packed in containers.  This will continue to be a high priority.
1. Community sampling 
4. The soil sampling plan has been revised to include screening for DEA.  The plan was approved today by LDEQ.  
4. Water, soil, and sediment samples have been obtained by the facility from the ditch adjacent to the site along Hwy 39 following cleanup activities conducted on 9/15/12.
1. CTEH conducting community air monitoring which continues to show no detects.
1. SPCC Plan still in development.
6. Professional Engineer is currently surveying containments and an SPCC Plan is being developed, repairs being made to containments as needed
1. Methyl Acrylate, Monochlorobenzene, and Formic acid scrubbers online and being monitored.
7. Scrubbers for Methyl Acrylate and Formic acid are currently being operated by air controlled pumps and portable air compressors.  Methyl Acrylate and Formic acid scrubber media were changed on 9/11/12.  Formic acid scrubber media were changed again today 9/17/12.
7. Visual checks of compressor and pump operation currently being conducted.  Spare compressors and pumps have been staged in case needed.
Completed actions:
1. 142 railcars have been re-railed as of 09/14/2012
1. All tank farm dikes have been reinforced to an appropriate level at this time. Ongoing reinforcement of walls will be addressed as noticed.
1. Styrene Tanks – (Inhibitor concentration and product temperatures are within appropriate/normal operating parameters.)  Portable chiller is currently being utilized.

